Author's response to reviews

Title: Arteriovenous malformation of the thyroid gland as a very rare cause of mechanical neck syndrome: a case report

Authors:

Monika Cerná (cernamo@fnplzen.cz)
Vladislav Treska (treska@fnplzen.cz)
Michal Krcma (krcmam@fnplzen.cz)
Ondrej Daum (daum@fnplzen.cz)
Frantisek Slauf (slauf@fnplzen.cz)

Version: 4
Date: 14 September 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

I am sending you our manuscript entitled „ Arteriovenous malformation of the thyroid gland as a rare cause of mechanical neck syndrom: a case report“ by Monika Černá et al.

We present interesting clinical case that describes symptoms, examination and approach to treatment of very rare localisation arteriovenous malformation in thyroid gland. We want to show non-specific symptoms that should mimic as a malignant disease preoperatively. It has been published a work in Spain that describes the same disease in two patient. We want to reach the worldwide literature with next case and demonstrate our original approach to treatment including endocrinologist, radiologist, pathologist and surgeon that has not been published so far.

Our case is an original case report that could be useful in differential diagnosis in lots of speciality of medicine and could be used as a manual for management to examination and treatment.

We summarize the available literature including the same disease. We want to emphasize the cooperation of specialist in more branch of medicine.

Please let me know of your decision at your earliest convenience.

With my best regards

Sincerely yours,

MUDr. Monika Černá, PhD.